Author's response to reviews

**Title:** Seroepidemiology of Toxocara Canis infection among primary schoolchildren in the capital area of the Republic of the Marshall Islands

**Authors:**

Chung-Jung Fu (subrina@shh.org.tw)
Ting-Wu Chuang (chtwingwu@tmu.edu.tw)
Huei-Shan Lin (13064@s.tmu.edu.tw)
Chih-Hsiung Wu (12340@s.tmu.edu.tw)
Yung-Ching Liu (yungchingliuster@gmail.com)
Mailynn K Langinlur (101652012@nccu.edu.tw)
Min-Yun Lu (13093@s.tmu.edu.tw)
Wesley Wei-Wen Hsiao (wesleyhsiao@hotmail.com)
Chia-Kwung Fan (tedfan@tmu.edu.tw)

**Version:** 4
**Date:** 29 April 2014

**Author's response to reviews:** see over
Reply to Reviewer and Editor

We are very grateful to your valuable comments on our manuscript and we have revised the ms according to both of your suggestions. Many thanks again!

Yours sincerely,

Corresponding author: Chia-Kwung Fan

#To Editor

Q. Please clarify participant consent.
A: We have added this declaration "...nurses collected blood specimen of PSC, after informed consent was obtained from PSC or parents/guardians, from schools..." in section of Study population and participant selection. Also, in Ethical approval part we added that "The research protocols were approved by the Institutional Review Board of Shuang Ho Hospital, Taipei Medical University (TMU-JIRB-No. 201306003) and also it was approved by Ministry of Health, RMI.".